Author’s response to reviews

Title: Vasorelaxant effects of Angelica decursiva root on isolated rat aortic rings

Authors:

Bumjung Kim (ori-pharm@hanmail.net)
Youngki Kwon (youngki0707@hanmail.net)
Somin Lee (sominlee@naver.com)
Kyungjin Lee (dostudy@naver.com)
Inhye Ham (iham@khu.ac.kr)
Ho Young Choi (hychoi@khu.ac.kr)

Version: 1 Date: 22 Mar 2017

Author’s response to reviews:

Dear Editor.

First of all, we wrote the reference about the compounds of Angelica decursiva at our manuscript. The title of reference is "Chemical Constituents and Anti-platelet Aggregation Activity from the Root of Peucedanum formosanum". In that paper, researchers compared many compounds such as imperatorin, bergepten which are included in Angelica decursiva and Peucedanum formosanum.

Also, regarding identification, we conducted gene analysis on Angelica decursiva in our manuscript. If you want further information of the phenolic content like HPLC data, please talk about your needs in detail.

sincerely yours.